Author’s response to reviews

Title: Syphilis-attributable adverse pregnancy outcomes in China: a retrospective cohort analysis of 1187 pregnant women with different syphilis treatment

Authors:
Huihui Liu (liuhh099@163.com)
Niannian Chen (cnnfezx@163.com)
Jia Yu (yjbigcs@163.com)
Weiming Tang (weimingtangscience@gmail.com)
Jianrong He (hjr0703@163.com)
Huiyun Xiao (xiaohuiyun1982@126.com)
Suifang Lin (lsfbigcs@163.com)
Fang Hu (hufang0351083@126.com)
Qiong Feng (fqfezx@163.com)
Joseph Tucker (jdtucker@med.unc.edu)
Huimin Xia (huimin.xia876001@gmail.com)
Xiu Qiu (qxiu0161@163.com)

Version: 1 Date: 17 May 2018

Author’s response to reviews:

Dear Editor,

Thank you for noticing us that our manuscript did not have information on informed consent of participants for the study. Now, we add it to the manuscript (line number, 85).

The PMID of references that were not be validated were followed: PMID: 27443780 of Reference 1, PMID:26981537 of Reference 9, PMID: 3679218 of Reference 14 and PMID: 26981537 of Reference 22.
Thank you for your consideration. We look forward to hearing from you soon.

Sincerely,

Xiu Qiu, MD, PhD